Author's response to reviews

Title: California Men's Health Study (CMHS): A multiethnic cohort in a managed care setting

Authors:

Shelley M Enger (shelleyenger@yahoo.com)
Barbara Sternfeld (Barbara.Sternfeld@kp.org)
Stephen K Van Den Eeden (Stephen.Vandeneeden@kp.org)
Ronald K Loo (Ronald.K.Loo@kp.org)
Charles P Quesenberry Jr (Charles.Quesenberry@kp.org)
Sarah Rowell (Sarah.Rowell@kp.org)
Marianne C Sadler (Marianne.C.Sadler@kp.org)
Donna M Schaffer (Donna.M.Schaffer@kp.org)
Laurel A Habel (Laurel.A.Habel@kp.org)
Bette J Caan (Bette.Caan@kp.org)

Version: 2 Date: 11 May 2006

Author's response to reviews:

Per the request of Emma Parkin, we are submitting a revised manuscript that removes sensitivity and specificity information, and adds a paragraph on the number of non-cancer cases we expect to accrue in a five year period following the baseline questionnaire. All other aspects of the manuscript and this submission are the same.